Author’s response to reviews

Title: An Investigation of Physical and Mental Health Consequences among Chinese Parents Who Lost Their Only Child

Authors:
qianlan Yin (yinqianlan@smmu.edu.cn)
Zhilei Shang (shangzhilei@163.com)
Na Zhou (nazhou@163.com)
liugang yu (liugangyu@smmu.edu.cn)
Lili Wu (wll_88@163.com)
Xiaoqian Yu (xiaoqian@mail.usf.edu)
Huaihui Zhang (huaihui_zhang@163.com)
Haidong Xue (amkeith@163.com)
Weizhi Liu (liuweizhi@smmu.edu.cn)

Version: 4 Date: 18 Jan 2018

Author’s response to reviews:

Thanks for the review and we have made all the needed adaption and correct the title page and author contribution. However, there is one important thing we must emphasize that from the beginning form of the manuscript, the order of the authors are the same. The difference between the submission system and the title page is the omissive filling of the form, and we have added all of the author in the submission system this time. We are sorry for the mistake, and we appreciate your understanding.